Author's response to reviews

Title: Selected acute phase CSF factors in ischemic stroke: findings and prognostic value

Authors:

Maia Beridze (postdoc2002@yahoo.com)
Tamar Sanikidze (sanikidze@tsmu.edu)
Roman R Shakarishvili (rshakarishvili@yahoo.com)
Nino Intskirveli (ssharashenidze@yahoo.de)
Natan M Bornstein (natanb@tasmc.health.gov.il)

Version: 5 Date: 3 March 2011

Author's response to reviews: see over
Dear Editor
Dear Editorial Board Members,

Once more thank you for taking in consideration the Manuscript: ID 5837172444441617, "Pathogenic Role and Prognostic Value of Selected Cerebrospinal Fluid Acute Phase Factors for Clinical Course and Outcome of Ischemic Stroke".

After the second revision of the manuscript the reviewer suggested to make some final changes and we fully accepted the notes he made.

Reviewer: Juergen Eggers
Minor compulsory revisions:

We considered the reviewer’s suggestions concerning the Table 1 and specified Group 1 and 2 in title.
In gender line we specified percent of males in Group 1 and 2 and abolished the line for females.
In age line we indicated (years) in parenthesis.

NIHSS we displayed as median (interquartil) range and explained this at the bottom of the Table 1.

There is no more criticism from both Reviewers.

We would like to express gratitude to reviewers for their notes that substantially improved the quality of manuscript.

All notes of Editorial Office are envisaged and changes are highlighted in the new version of manuscript.

With best regards,

Maia Beridze MD.PhD
Associated professor of Neurology
Department of Neuromedicine
Tbilisi State Medical University
Tbilisi, Georgia